Author's response to reviews

Title: Role of rapid urease test and histopathology in the diagnosis of Helicobacter pylori infection in a developing country

Authors:

Javed Yakoob (yakoobjaved@hotmail.com)
Wasim Jafri (wasim.jafri@aku.edu)
Shahab Abid (shahab.abid@aku.edu)
Nadim Jafri (nadijm@hotmail.com)
Zaigham Abbas (zaighamAbbbas@akunet.org)
Saeed Hamid (saeed.hamid@aku.edu)
Muhammad Islam (muhammad.islam@aku.edu)
Kashif Aniss (kashif.annis@aku.edu)
Hasnain AliShah (hasnain.Alishah@aku.edu)
Hizbullah Shaikh (hizbullah.shaikh@aku.edu)

Version: 4 Date: 11 October 2005

Author's response to reviews:

Title: Role of rapid urease test and histopathology in the diagnosis of Helicobacter pylori infection in a developing country

Reviewer: BJ Marshall

Thank you for your comments regarding our manuscript.
We have corrected the mistake in your initials in reference no: 9.

Reviewer: Max Bulsara
Thank you for your comments regarding our manuscript.
We agree with your comments.
1. We have changed the conclusion in the abstract to read "This study shows that the sensitivity, specificity, NPV and PPV of rapid urease test was reduced in patients who are on PPI".
2. We have added a sentence in the discussion page 8 lines 11-13, "Also, in view of the small sample size of our study, the result needs to be confirmed in a larger population of patients".